Author's response to reviews

Title: Implementing training and support, financial reimbursement, and referral to an internet-based brief advice program to improve the early identification of hazardous and harmful alcohol consumption in primary care (ODHIN): Study protocol for a cluster randomized factorial trial

Authors:

Myrna N Keurhorst (m.keurhorst@iq.umcn.nl)
Peter Anderson (peteranderson.mail@gmail.com)
Fredrik Spak (fredrik.spak@socmed.gu.se)
Preben Bendtsen (preben.bendtsen@liu.se)
Lidia Segura (lidia.segura@gencat.cat)
Joan Colom (joan.colom@gencat.cat)
Jillian Reynolds (reynolds@clinic.ub.es)
Colin Drummond (Colin.drummond@kcl.ac.uk)
Paolo Deluca (Paolo.deluca@kcl.ac.uk)
Ben van Steenkiste (ben.vansteenkiste@maastrichtuniversity.nl)
Artur Mierzecki (roklr@pum.edu.pl)
Karolina K#322;oda (wikarla@gazeta.pl)
Paul Wallace (p.wallace@ucl.ac.uk)
Dorothy Newbury-Birch (dorothy.newbury-birch@ncl.ac.uk)
Eileen Kaner (e.f.s.kaner@newcastle.ac.uk)
Toni Gual (tgual@clinic.ub.es)
Miranda GH Laurant (m.laurant@iq.umcn.nl)

Version: 3 Date: 20 January 2013

Author's response to reviews: see over
To:
The editor of “Implementation Science”

From:
M.N. Keurhorst
Scientific Institute for Quality of Healthcare (IQ healthcare)
Radboud University Nijmegen Medical Centre
P.O. Box 9101, 114 IQ healthcare
6500 HB Nijmegen
The Netherlands

Nijmegen, January 20, 2013

Dear editor,

Herewith, I would like to send you the carefully checked copy-edited version of our manuscript, titled “Implementing training and support, financial reimbursement, and referral to an internet-based brief advice program to improve the early identification of hazardous and harmful alcohol consumption in primary care (ODHIN): Study protocol for a cluster randomized factorial trial” for publication in the Implementation Science journal.

I accepted all of the changes that were made to conform the manuscript with the journal style. In addition, I would like to come back on some remarks of the editor. Remark JA1: I've indicated ‘*Corresponding author’ below my email address. Remark JA2: sentence is reformulated. Furthermore, I have made some other small changes: 1) ‘(ODHIN)’ added to the title; 2) As corresponded before, the number of authors has been decreased; 3) persons that are not authors anymore, are now acknowledged.

On behalf of all authors*, yours sincerely,

Mrs. Myrna Keurhorst, MSc

*Authors: